Author’s response to reviews

Title: Evaluation of Tp-e Interval, Tp-e/QT Ratio and Tp-e/QTc Ratio in Patients with Acute Myocarditis

Authors:

Fatih Mehmet Uçar (dr_fmucar@hotmail.com)
Cihan OZTURK (dr.cihanozturk@gmail.com)
Mustafa Adem YILMAZTEPE (mayilmaztepe@yahoo.com)

Version: 6 Date: 16 Sep 2019

Author’s response to reviews:

Dear Editor and reviewers,

Thank you for reading and reviewing our manuscript, which help us improve it to a better scientific level. We highly appreciated the constructive comments and valuable suggestions. We have modified the manuscript in response to reviewer comments and addressed each of their concerns point-by-point as detailed below.

Editor Comments:

1. Availability of Data and Materials

Please revise your current Availability of Data and Materials statement to reflect one of the formats indicated in our submission guidelines and clearly state who should be contacted if someone wants to request the data.: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

We suggest you use "The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.”.

2. Ethics approval and consent to participate

We note in your methods you state "All control group patients were healthy volunteers and study consent form was signed from all subjects.”

In your “ethical approval and consent to participate” section, please clarify whether informed consent, written or verbal, was obtained from all participants and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

3. Title Page

Please include a title page as the first page of your manuscript where you list the full names and
institutional addresses for all authors. The corresponding author should be indicated.

4. Copy Edit

Your manuscript needs to be copyedited. For example in the Study population section of your methods "approbated" should be "approved".

We recommend you proofread your manuscript and ensure that you correct any mistakes in your grammar. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may use a professional editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

5. Remove file

Please remove the Cover Letter from the file inventory as it is no longer needed at this stage of the editorial process.

6. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

RESPONSE: All parts were revised as your comments